Author's response to reviews

Title: Effects of Gyejibongnyeong-hwan on dysmenorrhea caused by blood stagnation: study protocol for a randomized controlled trial

Authors:

Jeong-Su Park (suyahpark@gmail.com)
Sunju Park (sjpark422@gmail.com)
Chun-Hoo Cheon (pm.thehoo@gmail.com)
Ho-Yeon Go (kohoyeon@gmail.com)
Seung-Ho Sun (sunguy2001@hanmail.net)
Yong-Cheol Shin (syc99@khu.ac.kr)
Seong-Gyu Ko (epiko@khu.ac.kr)

Version: 6 Date: 4 January 2012

Author's response to reviews: see over
January 4th, 2012

Re: Manuscript #1665788680553438

Dear Editor,

Thank you very much for your 4th revisional decision on our manuscript—Effects of Gyejibongnyeong-hwan on Dysmenorrhea caused by Blood Stagnation for Trials Journal (#1665788680553438). We have taken your critiques under consideration and revised the manuscript accordingly.

The revised version of the manuscript is included along with this letter for publication in Trials Journal.

Thank you very much for giving us a chance to publish our work in your journal.

Yours sincerely,

Seong-Gyu Ko, M.D., Ph.D., M.P.H.

Professor
Center for Clinical Research and Genomics,
Kyung Hee University.
1, Hoegi-dong, Dongdaemun-gu,
Seoul, 130-701, Korea

(Email) epiko@khu.ac.kr
(Phone) 82-2-961-0329
(Fax) 82-2-966-1165
Major revisions

1) Abstract: please ensure that the Abstracts in the submissions system and manuscript are identical.
   We ensure that the Abstracts in the submissions system and the manuscript are identical.

2) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
   We checked the manuscript and corrected some minor typographical errors.